Author's response to reviews

Title: Percutaneous injuries among dental professionals in Washington State

Authors:

Syed M Shah (syed.shah@usask.ca)
Anwar T Merchant (merchant@ccc.mcmaster.ca)
James A Dosman (james.dosman@usask.ca)

Version: 7 Date: 30 October 2006

Author's response to reviews: see over
October 30, 2006

Letter to the Editor.

We appreciate the final comments made by the BioMed Central Editorial Production Team. We have made the following changes as suggested by the Editorial Team.

1. Highlights were removed
2. Names of all co-authors added in ref. 10.

With best wishes;

Syed Shah, MD, MPH, PhD
Associate Professor, Department of Community Health and Epidemiology
University of Saskatchewan
The Canadian Centre for Health and Safety in Agriculture.
2734 Health Sciences Building
Wiggins Road, Saskatoon
S7N 5E5, SK
Canada
Email: syed.shah@usask.ca
Tel: 306-966-7977.